Author’s response to reviews

Title: What is the Difference in Morphologic Features of the Lumbar Vertebrae Between Caucasian and Taiwanese Subjects ? A CT-based Study : Implication of Pedicle Screws Placement via Roy-Camille or Weinstein Methods

Authors:

Hsi-Hsien Lin (hsihsienlin@gmail.com)
Jung-Pan Wang (jpwang0801@gmail.com)
Cheng-Li Lin (jengli94@gmail.com)
Shih-Tien Wang (stwang@vghtpe.gov.tw)
Ming-Chau Chang (mcchang@vghtpe.gov.tw)
Po-Hsin Chou (choupohsin@gmail.com)

Version: 2 Date: 04 Jan 2018

Author’s response to reviews:

Taipei Veterans General Hospital
No. 201, Sec. 2, Shipai Rd., Beitou District, Taipei City, Taiwan 11217, R.O.C.
Jan. 5, 2018

Dear Editors-in Chief Professor James Mockridge :

I am sending this manuscript entitled " The Lumbar Morphology and Pedicle Orientation Differ Between Caucasian and Taiwanese : Clinical Application " that we would like to submit for consideration to be published as an original paper in BMC Musculoskeletal Disorders.

We analyzed lumbar morphology and identical pedicle orientation at L1 to L5 based on CT scan in young patients (90 patients) and compared them with findings reported for Caucasian population. We used the previously reported techniques to quantitatively elucidate the parameters regarding lumbar morphology and identical pedicle orientation at each vertebrae.

All authors have read, approved, and affirmed the content of this submission for publication and the manuscript has not been published and is not being considered for publication elsewhere in whole or part in any language except as an abstract. We appreciate the
time and effort of your consideration of this material. Please direct all correspondence to the address below.

In consideration of BMC Musculoskeletal Disorders taking action in reviewing and editing my submission, the authors hereby transfer, assign, or otherwise convey all copyright ownership to BMC Musculoskeletal Disorders in the event that such work is published by BMC Musculoskeletal Disorders.

Thanks for everything.

Sincerely yours,

Po-Hsin Chou, M.D., PhD.

Department of Orthopedics & Traumatology, Taipei Veterans General Hospital
School of Medicine, National Yang-Ming University, Taipei, Taiwan, ROC